Author's response to reviews

Title: Primary repair of colon injuries: clinical study of non selective approach

Authors:

Ranko G Lazovic Dr (scp@beotel.yu)
Goran I Barisic Dr (barisic_goran@yahoo.com)
Zoran V Krivokapic Prof (scp@beotel.rs)

Version: 2 Date: 9 February 2010

Author’s response to reviews:

The paper was extensively revised as requested by reviewer. Native English speaker helped us to rewrite the paper. All spelling mistakes were corrected. You will see all changes in the uploaded document.